Author’s response to reviews

Title: Treatment of Urinary Calculi After Yang-Monti Ileal Ureter Reconstruction: A Case Report

Authors:

zhenxing wang (76166604@qq.com)
guangheng luo (13809440294@163.com)

ye tian (tianye5055@163.com)

xiushu yang (13985037039@139.com)

Version: 1 Date: 06 Jul 2018

Author’s response to reviews:

Dear Editor,

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Treatment of Urinary Calculi after Yang-Monti Ileal Ureter Reconstruction Using a Rigid Ureteroscope and a Flexible Ureteroscope”. Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

1. Written informed consent was obtained from the patient for publication of this Case Report and any accompanying images.

2. We have combined the Discussion and Conclusions section into an incorporated “Discussion and Conclusions” section in the main text of this manuscript.

3. We tried our best to improve the manuscript and made some changes in the manuscript.
We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval. Once again, thank you very much for your comments and suggestions.